Author's response to reviews

Title: Treatment-seeking Behaviour for Selected Reproductive Health Problems: Behaviours of Unmarried Female Adolescents for Selected Reproductive Health Problems in Two Low-performing Areas of Bangladesh

Authors:

Humayun Kabir (humayun@icddrb.org)
Nirod Chandra Saha (nirod@icddrb.org)
Andrea L Wirtz (awirtz1@jhu.edu)
Rukhsana Gazi (rukhsana@icddrb.org)

Version: 7 Date: 9 July 2014

Author's response to reviews:

July 09, 2014
To: Editor-in-Chief
The Reproductive Health Editorial Team
From: Humayun Kabir
Corresponding Author & Assistant Scientist
Centre for Equity and Health Systems, icddr,b
Dhaka, Bangladesh

Subject: Response to the review comments and suggestions in the revised manuscript “Treatment-seeking for Selected Reproductive Health Problems: Behaviours of Unmarried Female Adolescents in Two Low-performing Areas of Bangladesh” MS: 3695414911218998

Thank you for your review and suggestions. In response to these suggestions, this manuscript has undergone extensive revision and reorganization. Specifically, the content has been reorganized, shifting some material from the introduction and methods (limitations section) to the discussion. The discussion has been improved to adjust the order so that

1) The results of the study are summarized.
2) Strengths and weaknesses of the study are discussed.
3) Results of systematic reviews are discussed.
4) Policy implications are discussed.
5) Limitations and recommendations for improvements are presented.
6) Unanswered questions and future research are discussed.

The English language has been improved. We feel that the manuscript has
significantly improved and hope the reviewer agrees.

We are resubmitting herewith a revised version for your kind consideration.

Kind regards.

Humayun Kabir